Author's response to reviews

Title: The use of the Godin-Shephard Leisure-Time Physical Activity Questionnaire in oncology research: A systematic review

Authors:

Steve Amireault (steve.amireault@utoronto.ca)
Gaston Godin (gaston.godin@fsi.ulaval.ca)
Jason Lacombe (jason.lacombe@mail.utoronto.ca)
Catherine M Sabiston (catherine.sabiston@utoronto.ca)

Version: 4 Date: 18 June 2015

Author's response to reviews: see over
We thank the editor and the reviewers for their careful reading and constructive criticism of the original submission. We have made numerous changes to the manuscript on the basis of the reviews, and believe that the revised manuscript is considerably stronger as a result of these changes. Our responses (plain text) to each of the specific comments of the reviewers (italicized text) are described below. The revision and the ensuing modifications are identified in the text using a red font color.

1. We recommend that you copyedit the paper to improve the style of written English. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central recommends Edanz (www.edanzediting.com/bmc1). BioMed Central has negotiated a 10% discount to the fee charged to BioMed Central authors by Edanz. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication. For more information, see our FAQ on language editing services at http://www.biomedcentral.com/authors/authorfaq/editing.

Response 1: A native English speaker proofread the final version of this manuscript and the necessary corrections have been made. Please, note that the Limitations section (p. 17, lines 21-23 and p. 18, lines 1-12) have been reworked to improve clarity.

2. Please update the Competing Interest and Acknowledgment sections to remove the black boxes.

Response 2: This has been done.

3. Please include details of any funding in the Acknowledgment section.

Response 3: Details of any funding have been included in the Acknowledgment section (p. 19, lines 22-23).